Author's response to reviews

Title: Rapidly progressive dementia and Parkinsonism as a gliomatosis cerebri's initial clinical presentation in a 82 year old patient. A case report

Authors:

emmanuelle duron (emmanuelle.duron@brc.aphp.fr)
anne lazareth (lazarethanne@yahoo.fr)
jean-yves gaubert (jean-yves.gaubert@brc.aphp.fr)
carole raso (carole.raso@brc.aphp.fr)
olivier hanon (olivier.hanon@brc.aphp.fr)
anne-sophie rigaud (anne-sophie.rigaud@brc.aphp.fr)

Version: 4 Date: 16 November 2007

Author's response to reviews: see over
Dear Pr Kidd,

Please, find enclosed a manuscript by Dr Duron Emmanuelle et al. that we would like to resubmit as a Case Report entitled:

“Gliomatosis cerebri presenting as rapidly progressive dementia and Parkinsonism in an 82 year old patient. A case report”

Duron Emmanuelle MD, Lazareth A MD, Gaubert Jean-Yves MD, Raso Carole MD, Hanon O MD, PhD, Rigaud Anne-Sophie MD, PhD.

From: 1. Department of Geriatry, University Hospital Broca, AP-HP, France.

We confirm that the manuscript or part of the data have not been published or is not being simultaneously submitted elsewhere. All Authors have agreed to its submission.

The following revisions have been done as asked by the editorial board:
- The manuscript has been revised by a professional copyediting service, suggested by your journal

We hope that you will consider it suitable for publication.
Looking forward to hearing from you.

Sincerely yours,

Dr Emmanuelle Duron, MD